A Stackelberg Game Approach Towards Socially-Aware Incentive Mechanisms for Mobile Crowdsensing

Jiangtian Nie, Jun Luo, Senior Member, IEEE, Zehui Xiong, Dusit Niyato, Fellow, IEEE, Ping Wang, Senior Member, IEEE

Abstract-Mobile crowdsensing has shown a great potential to address large-scale data sensing problems by allocating sensing tasks to pervasive mobile users. The mobile users will participate in a crowdsensing platform if they can receive a satisfactory reward. In this paper, to recruit effectively and efficiently sufficient number of mobile users, i.e., participants, we investigate an optimal incentive mechanism of a crowdsensing service provider. We apply a two-stage Stackelberg game to analyze the participation level of the mobile users and the optimal incentive mechanism of the crowdsensing service provider using backward induction. In order to motivate the participants, the incentive mechanism is designed by taking into account the social network effects from the underlying mobile social domain. We derive the analytical expressions for the discriminatory incentive as well as the uniform incentive mechanisms. To fit into practical scenarios, we further formulate a Bayesian Stackelberg game with incomplete information to analyze the interaction between the crowdsensing service provider and mobile users, where the social structure information, i.e., the social network effects, is uncertain. The existence and uniqueness of the Bayesian Stackelberg equilibrium is analytically validated by identifying the best response strategies of the mobile users. Numerical results corroborate the fact that the network effects significantly stimulate higher mobile participation level and greater revenue of the crowdsensing service provider. In addition, the social structure information helps the crowdsensing service provider to achieve greater revenue gain.

Index Terms—Crowdsensing, social network effects, incentive mechanism, complete and incomplete information, Bayesian game, Stackelberg game, uncertainty, social influence

I. INTRODUCTION

In the past decade, we have been witnessing a fast proliferation of mobile users and devices in our daily lives. The ubiquitous mobile devices with various embedded functional sensors have remarkably promoted the information generation process. These advances stimulate the rapid development of mobile

Jiangtian Nie is with ERI@N, Interdisciplinary Graduate School as well as School of Computer Science and Engineering, Nanyang Technological University, Singapore. Jun Luo, Zehui Xiong, Dusit Niyato are with School of Computer Science and Engineering, Nanyang Technological University, Singapore. Ping Wang is with Department of Electrical Engineering and Computer Science, York University, Canada. sensing technologies, and mobile crowdsensing becomes one of the most attractive and popular paradigms. Mobile crowdsensing leverages the sensing capacity of worldwide available smart phones, e.g., GPS, camera and digital compass, to collect distributed sensory data.

A basic crowdsensing platform typically includes a cloudbased system and a collection of smart phones or mobile users. The platform can post a set of sensing tasks with different purposes, and mobile users are actively involved to perform the corresponding tasks. Realizing the great business potential, lots of crowdsensing-based applications have been designed and introduced in a number of areas. Sensorly [1] is dedicated for WiFi coverage information, Waze [2] and GreenGPS [3] are to collect road traffic information, DietSensor [4] is proposed to share and track users' diet and nutrition, and Noisetube [5] is for monitoring noise pollution.

Nevertheless, voluntary participation in the crowdsensing platform may not be sustainable. This is from the fact that the mobile users need to spend their own resources, e.g., smart phone battery, CPU computing power, storage memory, to accomplish the sensing tasks. Another major concern that discourages the mobile users from participation comes from the potential privacy issues. Therefore, individuals are reluctant to participate and share their collected information due to the lack of sufficient motivation and incentive. Nevertheless, the crowdsensing systems heavily rely on total user participation level and the individual contribution from each user. To stimulate and recruit users with mobile devices to participate in crowdsensing, the crowdsensing platform administrator, i.e., the Crowdsensing Service Provider (CSP), usually provides a reward for mobile users as a monetary incentive to compensate their cost or risk.

It is challenging to design the incentive mechanism that achieves a sustainable and profitable market for the CSP. When the reward is small, the collected sensing information from mobile participants is insufficient. Conversely, when the reward is large, the CSP may incur excessive operation cost. Accordingly, an efficient incentive mechanism has become an emerging topic of interest for a large number of researchers. However, most of the existing works have addressed the incentive mechanism for mobile crowdsensing without considering the interdependent behaviors of mobile users from social domain. This interdependency originates from network effects. Traditionally, *network effects* refer to the phenomenon that public goods or service is more valuable

This research is supported in part by the National Research Foundation, Prime Ministers Office, Singapore under its Energy NIC grant (NRF Award No.: NRF-ENIC-SERTD-SMES-NTUJTCI3C-2016), as well as by AcRF Tier 2 Grant MOE2016-T2-2-022. This work is also supported in part by WASP/NTU M4082187 (4080), Singapore MOE Tier 1 under Grant 2017-T1-002-007 RG122/17, MOE Tier 2 under Grant MOE2014-T2-2-015 ARC4/15, NRF2015-NRF-ISF001-2277, and EMA Energy Resilience under Grant NRF2017EWT-EP003-041. (*Corresponding author: Zehui Xiong.*)

if it is adopted by more users. In crowdsensing services, the participation behavior of mobile users can be deemed as buying "public goods", which means that the mobile users are more willing to participate if the number of other users is greater. For example, in a crowdsensing-based road traffic information sharing application, a user can receive a better and accurate traffic report if more users join and share their road information. Consequently, the complex and interdependent user behaviors post a remarkable challenge to the operation of the crowdsensing platform. More importantly, network effects frequently exist in densely connected social relationships, which is one of the key criteria to promote the wisdom of crowds [6], [7].

Nevertheless, only a few works [8], [9] have studied the incentive mechanism for crowdsensing and exploited network effects at the same time. The authors in [8], [9] investigated the behaviors of mobile users under global network effects¹, which is not appropriate for the structure of an underlying social domain. By contrast, social (local) network effects refer to the case where each user is only influenced directly by the decisions of other densely socially-connected users [11], [12]. For example, in a mobile crowdsensing platform for sharing road traffic information, a user (driver) can get a better route if more neighbourhood users (users in the same or nearby roads) of this user join and contribute their traffic data [2], [3]. On the contrary, this user cannot obtain any benefits if the users in other distant roads join and share their traffic data. This fact motivates us to explore the role of social (local) network effects in designing the incentive mechanism of crowdsensing services.

In this paper, we propose novel incentive mechanisms by leveraging the underlying social network effects to attract participants to crowdsensing platform. First, the crowdsensing platform administrator, i.e., the CSP, determines an incentive, i.e., the offered reward, to maximize its revenue. Then, based on the given reward, the mobile users decide on their participation level individually by taking the social network effects into account. The above rewarding and participating decision marking process can be inherently modeled as a hierarchical Stackelberg game. Moreover, we consider the uncertainty of social network effects, which commonly exists in the realworld crowdsensing applications. As such, we formulate the Bayesian Stackelberg game with incomplete information to analyze and evaluate the impacts of uncertainty of social network effects. The major contributions of this paper are summarized as follows:

• To the best of our knowledge, this is the first work on designing incentive mechanisms for mobile crowdsensing with the consideration of complete and incomplete information on social network effects. In particular, we incorporate the social network effects in the game model, which utilizes the structural properties from the underlying social domain, and fully characterizes the social relations among the mobile users.

- We model the interaction between the CSP and mobile users as a two-stage Stackelberg game and analyze each stage systematically through backward induction. We investigate two types of incentive mechanism for the crowdsensing platform with complete and incomplete information on social network effects, i.e., the Stackelberg game based incentive mechanism and the Bayesian Stackelberg game based incentive mechanism, respectively.
- In the Stackelberg game based incentive mechanism, the CSP and mobile users acquire the exact information on underlying social network effects. We propose the optimal incentive mechanism in terms of discriminatory incentive and uniform incentive, in which the CSP offers the different or the same reward to all the mobile users. For both, we are able to obtain the analytical expression for optimal reward.
- In the Bayesian Stackelberg game based incentive mechanism, the specific information on social network effects is under uncertainty. We obtain a unique Bayesian Nash equilibrium adopted by the mobile users in closed-form. Thereafter, the existence and uniqueness of the Bayesian Stackelberg equilibrium is proved by analyzing the best response strategies of the mobile users.
- Performance evaluation is provided to demonstrate the effectiveness of the proposed game theory based sociallyaware incentive mechanisms. Numerical results show that the network effects play an important role to promote higher participation level and thus greatly improve the revenue of the CSP. Moreover, the information about social relationship, i.e., social structure, helps the CSP to achieve greater revenue gain.

The rest of this paper is organized as follows. Section II provides the literature review. Section III describes the system model and the game formulation. In Section IV, we analyze the mobile user participation level and optimal reward using backward induction. In Section V, we formulate a Bayesian Stackelberg game where the social structure information is uncertain, and study the Bayesian game equilibrium. Section VI presents the performance evaluation and Section VII concludes the paper.

II. RELATED WORKS

Recently, a large number of prior works have been dedicated to designing incentive mechanisms [13]. Auction is a widelyadopted method to design the incentive mechanisms. In [14], the authors presented a mechanism for participation level determination and reward allocation using optimal reverse auction, in which the CSP receives service queries and initiates an auction for user participation. The authors in [15] explored the truthful mechanism with strong requirements of data integrity where the tasks are time window dependent. A reverse auction framework is adopted to derive the optimal incentive, which is computationally efficient and individually rational. In [16], the authors investigated scheduling problem, where the CSP announces a set of tasks and then mobile users compete for the tasks based on the sensing costs and available time periods. The approximation mechanisms for the CSP to schedule and

¹Global network effects refer to as the phenomenon that a user will obtain higher value when its behavior aligns with any other users [10].

reward the users under certain budget is provided. The authors in [17] studied incentivizing user participation and assigning location dependent tasks with capacity budget. A truthful oneround auction with approximation algorithm is proposed to obtain the optimal reward offered to the participants. In [18], the authors considered the user-centric model where each user can ask for reserve price, and designed the truthful and scalable auction mechanism for the CSP to achieve revenue maximization. The authors in [19] addressed how to maximize the valuation of the covered interested regions under limited budget for strategy-proof mobile crowdsensing. In [20], the authors proposed a long-term dynamic incentive mechanism to capture the dynamic nature of long-term data quality of participants, where a truthful, quality-aware and budget feasible algorithm is designed for task allocation with polynomial-time computational complexity. The authors in [21] investigated the auction based incentive mechanism considering social cost minimization and privacy preservation. The participants are selected based on predefined score functions by the CSP, and the computational efficiency, individual rationality, truthfulness and differential privacy are guaranteed. To prevent the Sybil attack where a user illicitly disguises other identities to obtain benefits, the authors in [22] designed Sybil-proof auction-based incentive mechanisms.

In addition to auction mechanism design, the incentive mechanisms are examined with different objectives. For example, the authors in [23] considered that the sensing information has an attached time-sensitive value that decreases over time and focused on the incentive design for cooperative data collection of participants. In [7], the authors explored the incentive mechanism with multiple CSPs, where the incentive mechanism is modeled as a noncooperative game. The discrete time dynamic inspired by the best response dynamics is proposed to achieve the Nash equilibrium of the modeled game. The authors in [24] presented a novel Vickrey-Clarke-Groves game based incentive mechanism for sensing resource sharing by the encouraged participants. The task allocation and resource sharing algorithm is developed to achieve the social fairness and efficiency tradeoff. The authors developed a new framework called Steered Crowdsensing in [25], which controls incentives by introducing gamifications with monetary reward to location-based services. In [26], the authors incorporated the consideration of data quality into the mechanism, and rewarded the participant depending on the quality of its collected data. The authors in [27] applied Tullock contests to design incentive mechanisms, where the reward includes a fixed contest prize, and Tullock prize function depending on the winner's contribution. In [28], the authors proposed a reward-based collaboration mechanism, where the CSP announces a total reward to be shared among collaborators, and the task and reward are allocated if sufficient number of participants are willing to collaborate. In [29], the authors studied a quality-aware Bayesian incentive problem for robust crowdsensing, where the data quality and sensing cost of users are drawn from known distribution.

In [30], the authors considered a sealed market for the CSP, where the participants have imperfect information on other participants behavior. The iterative game framework



Figure 1: Basic system model of mobile crowdsensing platform with social network effects.

is introduced and the incentive mechanism is obtained by best response dynamics with several iterations. The authors in [31] formulated the one-to-many Nash bargaining game to model the interaction between the CSP and participants. The distributed algorithm that ensures the participators' privacy and reduces the computation load of the CSP is provided. The authors in [32] proposed blockchain based distributed incentive mechanism which can remove the security threats caused by a "trustful" crowdsensing center. The participants with sensing information contribution obtain the reward that is recorded in transaction blocks. In our previous work [33], we considered the social network effects that promote the participation level while designing the incentive mechanism. In [34], the authors also highlighted the importance of "network effects" on social information sharing with the problem of dynamic routing. For example, a user traveling on one route benefits from the information collected by other users traveling on another route. However, the scenario where the network effects is certain has its limitation which may not be applicable to some of the real-world applications such as crowdsensing. In this paper, we consider the uncertain scenario where the social structure information is not known exactly by the CSP and participants.

III. SYSTEM DESCRIPTION AND GAME FORMULATION

We model the interaction among the Crowdsensing Service Provider (CSP) and the socially-aware participants, i.e., Mobile Users (MUs), as a hierarchical Stackelberg game, where the action of each MU is to choose an individual participation level and the action of the CSP is to give the payment as a reward to incentivize the MUs (Fig. 1). Consider a set of MUs denoted by $\mathcal{N} \triangleq \{1, \ldots, N\}$. Each MU $i \in \mathcal{N}$ determines its participation level or effort level, e.g., sensing data transmission frequency or sensing resolution, denoted by x_i where $x_i \in (0, +\infty)$.

Let $\mathbf{x} \stackrel{\Delta}{=} (x_1, \dots, x_N)$ and \mathbf{x}_{-i} denote the participation levels of all the MUs and all other MUs except MU *i*, respectively. The reward per effort unit provided to the MUs is given as: $\mathbf{r} = [r_1, \dots, r_i, \dots, r_N]^{\top}$. Then, the utility of MU *i* is given by

$$u_i(x_i, \mathbf{x}_{-i}) = f_i(x_i) + \Phi(x_i, \mathbf{x}_{-i}) + r(x_i) - c(x_i).$$
(1)

The first term $f_i(x)$ represents the private utility or internal effects that MU *i* obtains from the participation, which can

Table I: Main Notations

Symbol	Definition
\mathcal{N}, N	Set of MUs, and the total number of MUs, respectively
\mathcal{N}_i	Set of social neighbours of MU i
x_i	Participation level of MU <i>i</i> , i.e., the effort level in
	participation
$\mathbf{x}, \mathbf{x}_{-i}$	The participation levels of all the MUs and all other MUs
	except MU <i>i</i> , respectively
r_i	The offered reward to MU <i>i</i> from the CSP
a_i, b_i	The coefficients capturing the intrinsic value of different
	MUs
g_{ij}	The influence of MU j on MU i
c	The MU's unit cost associated to its participation level
μ	The parameter representing the equivalent monetary worth
	of MUs' participation level
s,t	The coefficients capturing the concavity of the profit
	obtained from the total contribution of all MUs
u_i	The utility of MU i
П	The revenue of the CSP
γ	Given social network effects coefficient
k	The out-degree of MU
l	The in-degree of MU
P(k)	The out-degree distribution of MUs
H(l)	The in-degree distribution of MUs
$\operatorname{Avg}(\mathbf{x}_{-i})$	The average participation level of social neighbours of
	MU i
\bar{k}	Average level of social network effects
σ_k^2, σ_l^2	The two variance of out-degree and in-degree distributions

be formulated as $f_i(x_i) = a_i x_i - b_i x_i^2$, where $a_i > 0$ and $b_i > 0$ are the coefficients that capture the intrinsic value of the participation to different MUs with heterogeneity [11], [12]. For example, in a crowdsensing-based traffic information sharing application, when a user reports speed and location on a certain road more frequently, i.e., larger x_i , the accuracy of the traffic condition on that road is higher [2], [3]. As in [11], the quadratic form of the internal utility not only allows for tractable analysis, but also serves as a good second-order approximation for a broad class of concave utility functions. Additionally, the linear-quadratic function captures the decreasing marginal returns from participation. In particular, a_i models the maximum internal participation willingness rate, and b_i is the willingness elasticity factor.

The second term, $\Phi(x_i, \mathbf{x}_{-i})$ denotes the external benefits gained from the network effects, which is the key component from Eq. (1). In crowdsensing applications, an MU can enjoy an additional benefit from information contributed or shared by the others [10]. The existing work explored the network effects of global nature, where the additional benefits due to new coming MUs are the same for all the existing MUs [8]. However, due to the structural properties from the underlying social domain, it is more appropriate to consider the network effects locally in crowdsensing service, i.e., the social network effects. Then, we introduce the adjacency matrix $\mathbf{G} = [g_{ij}]_{i,j \in \mathcal{N}}$. The elements in matrix g_{ij} indicates the influence of MU j on MU i, which can be unidirectional or bidirectional. For example, with a larger q_{ij} , the participation level of MU *j* can increase the utility of MU i faster. Motivated by the idea of social reciprocity [35], [36], a user's social behavior to another is likely to imitate the latter's behavior to the former, and vice versa. In other words, two friends in social network tend to influence each other and then behave similarly. As a result, the social ties from one MU to another MU, and vice versa tend to be the same. Thus, we consider $g_{ij} = g_{ji}$ in this paper, i.e., the social tie is reciprocal. Nevertheless, the proposed model can be applied to asymmetric social ties straightforwardly, interested readers can refer to Appendix C in [37]. Specifically, we adopt $\sum_{j \in \mathcal{N}} g_{ij} x_i x_j$ to represent the additional benefits obtained from the network effects, similar to that in [11], [12], [38].

The third term, $r(x_i)$, is the reward from CSP to the MU *i*, which is equal to $r_i x_i$, i.e., the reward is a linear function to the effort or participation level. The last term $c(x_i)$ denotes the cost associated to the participation level of the MU, e.g., energy consumption and network bandwidth consumed. Similar to [8], we assume that the cost is equal to cx_i , where *c* is the MU's unit cost. It is noted that, the same approach can be applied to the model with the heterogeneous unit cost (e.g., $c_i x_i^2$) straightforwardly, interested readers can refer to Appendix D in [37]. Then the utility of MU *i* is expressed by:

$$u_i(x_i, \mathbf{x}_{-i}, \mathbf{r}) = a_i x_i - b_i x_i^2 + \sum_{j=1}^N g_{ij} x_i x_j + r_i x_i - c x_i.$$
(2)

The monopoly CSP operates and maintains the platform with a fixed cost, which is ignored for the simplicity of the analysis later. Then, the formulation of revenue for the CSP is given by the payoff from total aggregated contribution of all MUs minus the total reward paid to MUs, i.e.,

$$\Pi = \mu \sum_{i=1}^{N} (sx_i - tx_i^2) - \sum_{i=1}^{N} r_i x_i.$$
 (3)

Similar to [12], we also use the linear-quadratic function for tractability to transform the MUs' participation level into the monetary revenue of the CSP, which features the law of diminishing return. That is, an MU's contribution increases with the MU's effort level but the marginal return decreases. μ is an adjustable parameter representing the equivalent monetary worth of MUs' participation level, and s, t > 0 are coefficients capturing the concavity of the function.

We first address the incentive mechanism by modeling the strategic interactions between the CSP and the MUs as a twostage single-leader multi-follower Stackelberg game.

Definition 1. Two-stage reward-participation game:

• Stage I (Reward): The CSP determines the reward, aiming at the highest revenue, i.e.,

$$\mathbf{r}^{*} = \arg \max_{\mathbf{r}} \left\{ \mu \sum_{i=1}^{N} (sx_{i} - tx_{i}^{2}) - \sum_{i=1}^{N} r_{i}x_{i} \right\};$$

Stage II (Participation): Each MU i ∈ N chooses the participation level x_i, given the observed reward r and the participation levels of other MUs x_{-i}, with the goal to maximize its individual utility, i.e.,

$$x_i^* = \arg\max_{x_i} u_i(x_i, \mathbf{x}_{-i}, \mathbf{r}).$$

We solve this two-stage Stackelberg game by finding a subgame perfect equilibrium for the cases of discriminatory incentive mechanism and uniform incentive mechanism for all MUs.

IV. STACKELBERG GAME EQUILIBRIUM ANALYSIS WITH COMPLETE INFORMATION

A. Stage II: MUs' participation equilibrium

Based on the definition of the Nash equilibrium, each MU chooses its participation level that is the best response. By setting the first-order derivative $\frac{\partial u_i(x_i,\mathbf{x}_{-i})}{\partial x_i}$ to 0, we obtain the best response of MU *i* as follows:

$$x_{i}^{*} = \max\left\{0, \frac{r_{i} - c + a_{i}}{2b_{i}} + \sum_{j=1}^{N} \frac{g_{ij}}{2b_{i}} x_{j}\right\}, \forall i.$$
(4)

Each MU's best response includes two parts. $\frac{r_i - c + a_i}{2b_i}$ is independent from the strategies of the other MUs, and $\sum_{j=1}^{N} \frac{g_{ij}}{2b_i} x_j$ is dependent on the other MUs' participation levels due to underlying social network effects. Although the participation level strategy of each MU is obtained as in Eq. (4), the Nash equilibrium cannot be ensured to be unique or even exist since each MU may unboundedly increase its participation level if the other MUs' participation levels are large enough. Therefore, we present a sufficient assumption, under which there exists the unique Nash equilibrium as described in Theorem 1. Regarding the assumption, the MU has the upper bound on participation level, e.g., due to the battery capacity of a mobile device, and thus Assumption 1 is reasonable.

Assumption 1.
$$\sum_{j=1}^{N} \frac{g_{ij}}{2b_i} < 1, \forall i.$$

Theorem 1. Under Assumption 1, the existence and uniqueness of MU participation equilibrium, i.e., the Nash equilibrium of Stage II in this Stackelberg game, can be guaranteed.

Proof. Existence of MU participation equilibrium: We denote \mathbf{x}^* as the strategy profile in the MU participation subgame, and x_i^{\dagger} as the largest participation level in \mathbf{x}^* . Then, we have

$$\begin{aligned} x_{i}^{\dagger} &= \left(\frac{r_{i} - c + a_{i}}{2b_{i}} + \sum_{j=1}^{N} \frac{g_{ij}}{2b_{i}} x_{j}\right)^{+} \\ &\leq \frac{r_{i} - c + a_{i}}{2b_{i}} + \sum_{j=1}^{N} x_{i}^{\dagger} \frac{g_{ij}}{2b_{i}} \\ &\leq \frac{|r_{i} - c + a_{i}|}{2b_{i}} + \sum_{j=1}^{N} x_{i}^{\dagger} \frac{|g_{ij}|}{2b_{i}}. \end{aligned}$$

Thus, under Assumption 1, we have $x_i^{\dagger} \leq \frac{|r_i - c + a_i|}{2b_i - \sum\limits_{i=1}^N |g_{ij}|} = \hat{x}$. As

a result, the strategy space $[0, \hat{x}]$ is convex and compact, and the utility function $u_i(x_i, \mathbf{x}_{-i})$ is continuous in x_i and \mathbf{x}_{-i} . We also have the second-order derivative of MU's objective function as follows: $\frac{\partial^2 u_i}{\partial^2 x_i} = -2b_i < 0$. Thus, the MU participation sub-game is a concave game which always admits the Nash equilibrium.

Uniqueness of MU participation equilibrium: Firstly, we have

$$-\frac{\partial^2 u_i}{\partial x_i^2} = -(-2b_i + g_{ii}) = 2b_i$$

Then, based on Assumption 1, we have

$$-\frac{\partial^2 u_i}{\partial x_i^2} > \sum_{j=1}^N g_{ij} = \sum_{j=1}^N |g_{ij}| = \sum_{j=1}^N \left| -\frac{\partial^2 u_i}{\partial x_i x_j} \right|, \quad (5)$$

which satisfies the dominance solvability condition, i.e., Moulin's Theorem [39]. As a result, the uniqueness of MU participation equilibrium is guaranteed under Assumption 1. The proof is then completed. \Box

Then, we propose the best response dynamics algorithm to obtain the Nash equilibrium with respect to MUs' participation level, as shown in Algorithm 1. The algorithm iteratively updates the MUs' strategies based on their best response functions in Eq. (4), and converges to the Nash equilibrium of MU participation sub-game.

Algorithm 1 Simultaneous best-response updating for finding Nash equilibrium of MU participation sub-game

1: Input:	
Precision threshold ϵ , $x_i^{[0]} \leftarrow 0$, $x_i^{[1]} \leftarrow 1 + \epsilon$, $k \leftarrow 1$;	
2: while $\ x_i^{[k]} - x_i^{[k-1]}\ _1 > \epsilon$ do	
3: for all $i \in \mathcal{N}$ do	
4: $x_i^{[k+1]} = \left(\frac{r_i - c + a_i}{2b_i} + \sum_{j=1}^N x_j^{[k]} \frac{g_{ij}}{2b_i}\right)^+;$	
5: end for	
6: $k \leftarrow k+1;$	
7: end while	
8: Return $\mathbf{x}_i^{[K]}$;	

Proposition 1. Algorithm 1 achieves the Nash equilibrium of MU participation sub-game.

Note that Algorithm 1 achieves the approximate Nash equilibrium of MU participation sub-game, and the approximate accuracy, which measured by the gap between the achieved results and the actual solutions, depends on the precision threshold ϵ . The convergence speed of the proposed algorithm also depends on precision threshold ϵ . When ϵ is small, the number of iterations needed is large but the achieved results are more accurate. Conversely, when ϵ is big, the number of iterations needed is small but the achieved results are less accurate.

For ease of presentation, we have the following definitions, $\mathbf{B} := diag(2b_1, 2b_2, \dots, 2b_N)$, $\mathbf{a} := [a_i]_{N \times 1}$, $\mathbf{1} := [1]_{N \times 1}$, $\mathbf{G} := [g_{ij}]_{N \times N}$, $\mathbf{r} := [r_i]_{N \times 1}$ and \mathbf{I} is an $N \times N$ identity matrix. For the rest of the paper, similar to [12], [40], we consider the practical situation where all the MUs have positive participation levels at the Stackelberg equilibrium, i.e., a special case of Eq. (4). Then, with Lemma 1, we can rewrite Eq. (4) in a matrix form as follows:

$$\mathbf{x} = \mathbf{K} \left(\mathbf{a} + \mathbf{r} - c\mathbf{1} \right), \tag{6}$$

where $\mathbf{K} = (\mathbf{B} - \mathbf{G})^{-1}$.

Lemma 1. $\mathbf{B} - \mathbf{G}$ is positive definite matrix, which is invert*ible*.

Proof. We first denote $(\mathbf{B} - \mathbf{G})_{ij}$ as the value in the *i*th row and the *j*th column of the matrix $\mathbf{B} - \mathbf{G}$, and it holds that $(\mathbf{B} - \mathbf{G})_{ii} = 2b_i - g_{ii} = 2b_i$ since we have $g_{ii} = 0$. Under

Assumption 1, we also have $2b_i > \sum_{i=1}^N g_{ij}$. Furthermore, we

observe that $\sum_{j=1}^{N} g_{ij} = -\sum_{j \neq i} (\mathbf{B} - \mathbf{G})_{ij} = \sum_{j \neq i} \left| (\mathbf{B} - \mathbf{G})_{ij} \right|.$ Therefore, it holds that $(\mathbf{B} - \mathbf{G})_{ii} = 2b_i > \sum_{j \neq i} \left| (\mathbf{B} - \mathbf{G})_{ij} \right|.$

Accordingly, $\mathbf{B} - \mathbf{G}$ is strictly diagonally dominant and all the diagonal elements, i.e., $2b_i$ are larger than 0. Based on Gershgorin circle theorem [41], every eigenvalue λ of **B** – **G** satisfies

$$\left| \left(\mathbf{B} - \mathbf{G} \right)_{ii} - \lambda \right| < \sum_{j=1}^{N} \left| \left(\mathbf{B} - \mathbf{G} \right)_{ij} \right|.$$
 (7)

Moreover, we know $\lambda > 0$, and thus **B** – **G** is a positive definite matrix, from which its invertibility follows. The proof is then completed.

B. Stage I: Optimal incentive mechanism

In this stage, the monopoly CSP determines the reward to be paid to the MUs, the objective of which is to maximize the CSP's revenue. Specifically, we investigate the discriminatory incentive mechanism and the uniform incentive mechanism, respectively. The significance of each incentive mechanism is as follows. Under the uniform incentive mechanism, the equilibrium ensures a fair reward applied to all MUs. Moreover, the uniform incentive mechanism is simple to implement in the crowdsensing applications. However, the CSP has limited degree of freedom to maximize its profit. By contrast, under the discriminatory incentive mechanism, the CSP can customize the reward for each MU, matching with the MU's preference and capability. As such, the profit obtained under the discriminatory incentive mechanism is expected to be higher than that of the uniform incentive mechanism. This is also confirmed in our numerical results.

1) Discriminatory incentive mechanism: Under reward discrimination, the CSP is able to provide different reward for different MUs as incentive to maximize its revenue. The revenue maximization problem can be formulated as follows:

maximize
$$\Pi = \mu \sum_{i=1}^{N} (sx_i - tx_i^2) - \sum_{i=1}^{N} r_i x_i$$
$$= \mu (s\mathbf{1}^\top \mathbf{x} - \mathbf{x}^\top t\mathbf{x}) - \mathbf{r}^\top \mathbf{x}.$$
(8)

subject to $\mathbf{x} = \mathbf{K} (\mathbf{a} + \mathbf{r} - c\mathbf{1})$.

By plugging into the objective funcх tion in Eq. (8), we have $\Pi = \mu \left(s \mathbf{1}^{\top} \mathbf{K} \left(\mathbf{a} + \mathbf{r} - c \mathbf{1} \right) - t \left(\mathbf{a} + \mathbf{r} - c \mathbf{1} \right)^{\top} \mathbf{K}^{2} \left(\mathbf{a} + \mathbf{r} - c \mathbf{1} \right) \right) \mathbf{r}^{\top}\mathbf{K}(\mathbf{a}+\mathbf{r}-c\mathbf{1})$. Taking the partial derivative of the objective function in Eq. (8) with respect to the decision vector **r** to zero, i.e., $\frac{\partial \Pi}{\partial \mathbf{r}} = 0$, we obtain $\mu \left(s\mathbf{K}\mathbf{1} - 2t\mathbf{K}^2 \left(\mathbf{a} + \mathbf{r} - c\mathbf{1} \right) \right) - \mathbf{K} \left(\mathbf{a} + \mathbf{r} - c\mathbf{1} \right) - \mathbf{Kr} = 0$. Then, we have $\mu \left(s\mathbf{K1} - 2t\mathbf{K}^2 \left(\mathbf{a} - c\mathbf{1} \right) \right) - \mathbf{K} \left(\mathbf{a} - c\mathbf{1} \right) =$ $(2\mathbf{K} + 2\mu t\mathbf{K}^2)\mathbf{r}$. Finally, we obtain the optimal value \mathbf{r}^* , which is represented as follows:

$$\mathbf{r}^* = (2\mathbf{I} + 2\mu t\mathbf{K})^{-1} \left(\mu \left(s\mathbf{1} - 2t\mathbf{K} \left(\mathbf{a} - c\mathbf{1}\right)\right) - \left(\mathbf{a} - c\mathbf{1}\right)\right).$$
(9)

2) Uniform incentive mechanism: In this case, the CSP can only choose a single uniform reward to be paid to all the MUs, i.e., $r_i = r$, for all *i*. Then, the optimization problem is given by

maximize
$$\Pi = \mu \sum_{i=1}^{N} (sx_i - tx_i^2) - r \sum_{i=1}^{N} x_i$$
$$= \mu (s\mathbf{1}^{\top}\mathbf{x} - \mathbf{x}^{\top}t\mathbf{x}) - r\mathbf{1}^{\top}\mathbf{x}.$$
(10)
subject to $\mathbf{x} = \mathbf{K} [\mathbf{a} + (r-c)\mathbf{1}].$

Similarly, we eliminate x from the objective function in Eq. (10), and we obtain $\Pi = \mu(s\mathbf{1}^{\top}\mathbf{K}(\mathbf{a}+(r-c)\mathbf{1})$ $t(\mathbf{a} + (r-c)\mathbf{1})^{\top}\mathbf{K}^{2}(\mathbf{a} + (r-c)\mathbf{1})) - r\mathbf{1}^{\top}\mathbf{K}(\mathbf{a} + (r-c)\mathbf{1}).$ Then, we evaluate its first-order optimality condition to the reward r, which yields $\mu \left(s \mathbf{1}^{\top} \mathbf{K} \mathbf{1} - 2t (\mathbf{a} + (r-c)\mathbf{1})^{\top} \mathbf{K}^2 \mathbf{1} \right)$ – with respect $rac{\partial \Pi}{\partial r}$ = $\mathbf{1}^{\top}\mathbf{K}(\mathbf{a}+(r-c)\mathbf{1}) - r\mathbf{1}^{\top}\mathbf{K}\mathbf{1} = 0$. As a result, with simple steps, we obtain the optimal value of the uniform reward, which is represented by

s

$$r^* = \left(2\mu t \mathbf{1}^{\top} \mathbf{K}^2 \mathbf{1} + 2\mathbf{1}^{\top} \mathbf{K}\right)^{-1} \left\{ \mu \left[s \mathbf{1}^{\top} \mathbf{K} \mathbf{1} - 2t (\mathbf{a} - c \mathbf{1})^{\top} \right. \\ \times \mathbf{K}^2 \mathbf{1} \right] - \mathbf{1}^{\top} \mathbf{K} \left(\mathbf{a} - c \mathbf{1} \right) \right\}.$$
(11)

Until now, we have obtained the optimal incentive mechanism in terms of uniform reward and discriminatory reward in a closed-form solution with complete information, and hence validated the uniqueness of the Stackelberg equilibrium.

V. BAYESIAN STACKELBERG GAME THEORETIC ANALYSIS FOR SOCIALLY-AWARE INCENTIVE MECHANISM WITH INCOMPLETE INFORMATION

Recall from Section IV, we assume that the MUs will truthfully report their personal information (type) to the CSP. This situation can happen when there exists a supervising entity in the market that is capable of monitoring, sharing and storing all behaviors to ensure that the MUs always report the correct information. However, without the supervising entity which is often the case in practice, the MU does not reveal private information (type) to the CSP because of the concern on privacy leakage or selfish behaviors. Therefore, the incomplete information scenario is more suitable and applicable to the real-world crowdsensing applications and address the incentive mechanism therein. In this section, we extend the analysis to the scenario where the social structure information, i.e., the social network effects, is not exactly known by the CSP and MUs. Thus, we formulate the incentive mechanism as a Bayesian Stackelberg game [42], [43], and evaluate the game equilibrium by defining and optimizing the expected utility of MUs and the expected revenue of the CSP.

A. Problem formulation with social structure uncertainty

In the model proposed in Section III, the important social structure information may be uncertain or unknown by the decision makers, i.e., the CSP and MUs. Accordingly, this game can be modeled as a Bayesian game where the Bayesian analysis is adopted to predict the game outcome. In particular, the social relationship, i.e., social structure of each MU is private information and is considered as the type of the followers. Only its probability distribution is commonly known. Such distribution information can be obtained through, e.g., historical information or long-term learning.

The mobile social structure is represented by an interaction matrix, i.e., the adjacency matrix **G**. As aforementioned, the element g_{ij} denotes the strength of the influence of MU j on MU i. Recall from Section III, the utility of an MU can be expressed as follows:

$$u_i(x_i, \mathbf{x}_{-i}, \mathbf{r}) = x_i - \frac{1}{2}{x_i}^2 + \sum_{j=1}^N g_{ij} x_i x_j + r_i x_i - c x_i.$$
(12)

Note that we set $a_i = 1$ and $b_i = 1/2$ in order to concentrate on the social structure uncertainty. Moreover, without loss of generality, for all the social neighbours of MU *i*, i.e., $j \in \mathcal{N}_i$, $g_{ij} = \gamma > 0$, and γ is a given social network effect coefficient. Thus, the above equation is rewritten as follows:

$$u_i(x_i, \mathbf{x}_{-i}, \mathbf{r}) = x_i - \frac{1}{2}{x_i}^2 + \gamma x_i \sum_{j \in \mathcal{N}_i} x_j + r_i x_i - c x_i.$$
(13)

Therefore, the expected utility is expressed as follows:

$$U_{i}(x_{i}, \mathbf{x}_{-i}, \mathbf{r}) = E\left[u_{i}(x_{i}, \mathbf{x}_{-i}, \mathbf{r})\right] = x_{i} - \frac{1}{2}{x_{i}}^{2}$$
$$+ \gamma x_{i} E\left[\sum_{j \in \mathcal{N}_{i}} x_{j}\right] + r_{i} x_{i} - c x_{i}. \quad (14)$$

The social structure leads to different in-degrees and outdegrees of MUs. The in-degree denotes the number of other MUs that a certain MU influences, the out-degree denotes the number of other MUs influencing this MU. Thus, the indegree represents its influence and the out-degree represents its susceptibility. The distribution² of in-degree and out-degree captures the social network effects from the network interaction patterns [11], [44]–[47]. Note that the proposed model can still be applied to the asymmetric social ties, since we consider both the in-degree and out-degree distributions of each MU instead of the degree distribution. For example, an MU Alice has the social influence on MU Bob, but Bob may not have the social influence on Alice. The reason is that Alice may have different in-degree and out-degree.

The in-degree $l \in D$ and out-degree $k \in D$, where $D = \{0, 1, \ldots, k^{max}\}$ and k^{max} denotes the maximum possible value. We define $P: D \to [0, 1]$ and $H: D \to [0, 1]$ as the probability distributions of out-degree and in-degree, respectively, and we have $\sum_{k \in D} P(k) = \sum_{l \in D} H(l) = 1$. Furthermore, we assume that two probability distributions are independent and their variances are denoted as σ_k^2 and σ_l^2 , respectively. Due to consistency theory, we know

 $\sum_{k \in D} P(k)k = \sum_{l \in D} H(l)l = \overline{k}$, and thus \overline{k} is referred to as the mean value of social network effects. Moreover, we have

$$\mathbb{E}\left[\sum_{j\in\mathcal{N}_i} x_j\right] = k_i \times \operatorname{Avg}(\mathbf{x}_{-i}), \tag{15}$$

where $\operatorname{Avg}(\mathbf{x}_{-i}) = \mathbb{E}[x_j | j \in \mathcal{N}_i]$ denotes the average participation level of social neighbours of MU *i*.

In order to obtain the expression of $\operatorname{Avg}(\mathbf{x}_{-i})$, we employ the concept of "Configuration Model" in Network Science [48] to model the random networks generated with only in-degree distribution. According to Configuration Model's property (See Chapter 12.2 in [48]), to a user, the degree distribution of its randomly chosen neighbor is $\overline{H}(l) = \frac{H(l)l}{\sum\limits_{i'\in D} H(l')l'}$. In other words, a randomly selected social neighbours of MU *i* has the in-degree distribution as $\overline{H}(l)$ and out-degree distribution as P(k). Thus, by denoting the participation level of the MU with out-degree *k* and in-degree *l* as x(k, l), we have [46], [47]

$$\operatorname{Avg}(\mathbf{x}_{-i}) = \sum_{l \in D} \bar{H}(l) \left(\sum_{k \in D} P(k) x(k, l) \right), \quad (16)$$

where $\overline{H}(l) = \frac{H(l)l}{\sum\limits_{l'\in D} H(l')l'}$. Note that given $\operatorname{Avg}(\mathbf{x}_{-i})$, the participation level of MU *i* only depends on the reward and its out-degree *k*. Thus, the final expected utility of MU *i* is expressed as follows:

$$U_{i}(x_{i}, \mathbf{x}_{-i}, \mathbf{r}, k_{i}) = (1 + r_{i} - c)x_{i} - \frac{1}{2}x_{i}^{2} + \gamma k_{i}x_{i}\operatorname{Avg}(\mathbf{x}_{-i}),$$
(17)

and the type of the MU is its in-degree and out-degree, which is denoted as (l, k).

Since only the distribution of the in-degree and out-degree information is known, instead of maximizing the revenue as defined in Eq. (3), the objective of the leader, i.e., the CSP, is to maximize its expected revenue, which is given as follows:

$$\Pi = \sum_{l \in D} \left(\sum_{k \in D} H(l) P(k) \left((\mu s - r(k, l)) x(k, l) - \mu t x^2(k, l) \right) \right), \quad (18)$$

where r(k, l) is the reward offered to the MU with out-degree k and in-degree l.

B. Stackelberg game equilibrium analysis

We also adopt the backward induction to analyze the Bayesian Stackelberg game.

1) Follower game: For the given incentive or the reward determined by the CSP, we examine the Bayesain Nash equilibrium in the follower game which is characterized by the following theorem.

Theorem 2. The existence and uniqueness of the Bayesian Nash equilibrium of the follower game can be guaranteed, provided that the following condition

$$\gamma k^{\max} < 1 \tag{19}$$

holds.

²Since the social ties/links are constructed by the in/out-degree information of MUs, the social ties/links are also treated as random variables in some sense. Note that γ in the model is a given social network effect coefficient, which captures the strength of social ties/links. Although the value of γ is given, the social ties/links follow certain probability distribution instead of being a constant value. It is also noteworthy that γ can be treated as an approximate term instead of an exact value. Nevertheless, the impacts of uncertainty of γ can still be absorbed into the distribution of in/out-degree since these impacts are interdependent.

Proof. The existence of Bayesain follower game: To prove that there exists at least one Bayesian Nash equilibrium in the follower game (Proposition 1 in [49]), we need to ensure that the following condition

$$\frac{\partial U_i(\overline{x}, \mathbf{x}_{-i}, \mathbf{r}, k_i)}{\partial x_i} \le 0,$$

$$\forall i \in \mathcal{N}, k \in Z^+, r \in \mathbb{R}^+, \exists \overline{x} \ge 0, \forall x \le \overline{x} \quad (20)$$

holds, where $\overline{x} = \operatorname{Avg}(\mathbf{x}_{-i})$. Since we have

$$\frac{\partial U_i(\bar{x}, \mathbf{x}_{-i}, \mathbf{r}, k_i)}{\partial x_i} = (1 + r_i - c) - \bar{x} + \gamma k_i \operatorname{Avg}(\mathbf{x}_{-i})$$

$$\leq (1 + r_i - c) - \bar{x} + \gamma k^{\max} \bar{x}$$

$$= 1 + r_i - c + (\gamma k^{\max} - 1) \bar{x}, \quad (21)$$

we can ensure that the condition in Eq. (20) holds provided that $\gamma k^{\max} < 1$ is satisfied.

The uniqueness of Bayesain follower game: The proof of the uniqueness of the pure Bayesian Nash equilibrium can be directly derived from [49]. In particular, the sufficient condition that implies there exists at most one Bayesian Nash equilibrium is given as follows (Proposition 3 in [49]):

$$\left|\frac{\partial^2 U_i(\bar{x}, \mathbf{x}_{-i}, \mathbf{r}, k_i)}{\partial x_i \partial \operatorname{Avg}(\mathbf{x}_{-i})} \middle| \frac{\partial^2 U_i(\bar{x}, \mathbf{x}_{-i}, \mathbf{r}, k_i)}{\partial x_i \partial x_i} \right| < 1, \forall i \in \mathcal{N}.$$
(22)

With simple steps, we have $\left|\frac{\partial^2 U_i(\bar{x}, \mathbf{x}_{-i}, \mathbf{r}, k_i)}{\partial x_i \partial \operatorname{Avg}(\mathbf{x}_{-i})} / \frac{\partial^2 U_i(\bar{x}, \mathbf{x}_{-i}, \mathbf{r}, k_i)}{\partial x_i \partial x_i} \right| = |\gamma k_i| \leq |\gamma k^{\max}|.$ Thus, if $\gamma k^{\max} < 1$ holds, the condition given in Eq. (22) is guaranteed. The proof is then completed.

To obtain the closed-form expression of the unique Bayesian Nash equilibrium point in the follower game, we first apply partial derivative of the expected utility given in Eq. (17), i.e., $\frac{\partial U_i(\bar{x}, \mathbf{x}_{-i}, \mathbf{r}, \mathbf{k}_i)}{\partial x_i} = 0$, as shown as follows:

$$x_i^* = 1 + r_i - c + \gamma k_i \mathbb{E} \left[x_j | j \in \mathcal{N}_i \right].$$
(23)

Thus, we have

$$x(k,l) = 1 + r(k,l) - c + \gamma k \mathbb{E} \left[x(k,l) \, \big| (k,l) \in D^2 \right].$$
(24)

From Eq. (16), we have

$$\mathbb{E}\left[x(k',l') \left| (k',l') \in D^2\right] = \sum_{l' \in D} \overline{H}(l') \sum_{k' \in D} P(k')x(k',l') \\ = \sum_{l' \in D} \left(\overline{H}(l') \sum_{k' \in D} (P(k')(1+r(k',l')-c) \\ + \gamma k' \mathbb{E}\left[x(k'',l'') \left| (k'',l'') \in D^2\right]\right)\right) \\ = 1 + \overline{r} - c + \gamma \overline{k} \mathbb{E}\left[x(k'',l'') \left| (k'',l'') \in D^2\right], \quad (25)$$

where $\overline{r} = \sum_{l \in D} \overline{H}(l) \sum_{k \in D} P(k)r(k,l)$ and $\overline{k} = \sum_{l \in D} \overline{H}(l) \sum_{k \in D} P(k)k = \sum_{k \in D} P(k)k$. Since we also have $\mathbb{E} \left[x(k',l') \left| (k',l') \in D^2 \right] = \mathbb{E} \left[x(k'',l'') \left| (k'',l'') \in D^2 \right] \right],$ (26)

it can be concluded from Eq. (25) with the following expression 1 + -

$$\operatorname{Avg}(\mathbf{x}_{-i}) = \mathbb{E}\left[x_j \mid j \in N_i\right] = \frac{1+r-c}{1-\gamma \overline{k}}.$$
 (27)

Therefore, we obtain the closed-form expression of the participation level of the MU with type (k, l) in the Bayesian follower game, which is given as follows:

$$x^{*}(k,l) = 1 + r(k,l) - c + \gamma k \frac{1 + \overline{r} - c}{1 - \gamma \overline{k}}.$$
 (28)

Note that Algorithm 1 can be implemented similarly in incomplete information scenario. The only difference is that the best-response function update policy in Line 4 of Algorithm 1 is replaced by another update policy obtained from Eq. (19). Since we have validated the existence and uniqueness of the Bayesian Nash equilibrium, the modified Algorithm 1 can achieve the Bayesian Nash equilibrium [50]. Similar to that in the complete information scenario, this Bayesian Nash equilibrium is also the approximate equilibrium due to the error ϵ .

2) Leader game: As the CSP has the information on the degree distributions of MUs but has no information on MUs' type, and thus can offer only a uniform reward³, i.e., r(k, l) = r for all MUs. The optimal incentive mechanism obtained from the leader game is characterized by the following theorem.

Theorem 3. The optimal reward offered by the CSP in the Bayesian Stackelberg game is unique, which is given as follows:

$$r^* = c - 1 + \frac{(\mu s + 1 - c) \left(1 - \gamma \overline{k}\right)}{2 \left(1 - \gamma \overline{k} + \mu t + \mu t \gamma^2 \sigma_k^2\right)}.$$
 (29)

Proof. Please refer to the appendix for the details. \Box

Furthermore, we study the benchmark case where the CSP knows both the in-degree and out-degree of any individual follower. In such a situation, the CSP is able to offer discriminatory reward as incentive, r(k, l) for the MU with out-degree k and in-degree l. Then, the revenue maximization problem faced by the CSP is characterized in the following theorem.

Theorem 4. Provided that the CSP clearly knows the type of each individual MU, the optimal discriminatory reward r(k, l) offered to the MU with out-degree k and in-degree l, is unique.

Proof. Please refer to the appendix for the details. \Box

VI. PERFORMANCE EVALUATION

In this section, we evaluate the performance of the proposed socially-aware incentive mechanisms of the CSP in crowdsensing applications, and investigate the impacts of different parameters of mobile networks on the performance.

³Note that the uniform incentive mechanism is more applicable in incomplete information scenario, where the CSP has no information on the specific type of each individual MU. However, the in/out-degree distributions of MUs can be obtained through, e.g., historical information or long-term learning, which makes the uniform incentive mechanism feasible. This can also be confirmed by the closed-form solution for the optimal uniform reward, since the expression of the optimal reward only includes the mean and variance of the in/out-degree distributions instead of the type of individual MU.



Figure 2: The impact of total number of MUs on the crowdsensing service provider and mobile participants.

A. Investigation on Stackelberg game with complete information on social structure

We consider a group of N MUs, i.e., mobile participants, in a social network and set the parameters as follows. We assume the intrinsic parameters of MUs, i.e., a_i and b_i follow the normal distribution $\mathcal{N}(\mu_a, 2)$ and $\mathcal{N}(\mu_b, 2)$. In addition, the social tie g_{ij} between any two users *i* and *j* follows a normal distribution $\mathcal{N}(\mu_g, 1)$. The default parameters are set as: c = 15, $\mu = 0.1$, s = 20, t = 0.05, $\mu_a = \mu_b = 15$, $\mu_g =$ 0.05 and N = 50. Note that some of these parameters are varied according to the evaluation scenarios. As expected and verified in Fig. 2 and Fig. 3, the discriminatory incentive yields the larger revenue for the CSP, compared with the uniform incentive. Intuitively, the reason is that the CSP can adjust the reward according to individual MU's effort and contribution, which is proven by Fig. 4.

We next evaluate the impact of the total number of MUs on the proposed incentive mechanisms, as illustrated in Fig. 2. As the number of MUs increases, the total utilities of participants and the revenue of the CSP also increase under both mechanisms. The reason is that when the total number of MUs increases, the number of social neighboring MUs also increases. Owing to the underlying social network effects, the MUs are motivated by their social neighbours to have higher participation levels, and the revenue of the CSP is improved accordingly. In addition, with the increase of total number of participants, the total offered reward increases since the CSP tends to encourage more MUs to participate, in order to attain a greater revenue gain. In particular, the discriminatory and uniform incentive mechanisms enable the CSP to reduce the reward paid to the MUs, i.e., the cost, and therefore achieve a greater revenue gain in turn. Figure 3 depicts the impact of average value of social network effects on two entities of this network, i.e., the CSP and MUs. We observe that as the social network effects becomes stronger, the total utilities of



Figure 3: The impact of average value of social network effects on the crowdsensing service provider and mobile participants.

participants and the revenue of the CSP also increase. Since when the strength of social tie is stronger, the additional benefits obtained from social network effects are greater. In other words, the socially-aware MUs are motivated by each other and have higher participation levels consequently. When the participation levels are high enough, the CSP is able to offer less reward to save money. In turn, the total utilities of participants and the revenue of the CSP are improved. Furthermore, we observe that the total offered reward under the uniform incentive mechanism and the discriminatory incentive mechanism have no big difference from both Figs. 2 and 3. The reason is that the CSP under the discriminatory incentive mechanism is able to achieve a balanced reward allocation with the similar cost. For example, the CSP can offer more reward to some MUs and less reward to some other MUs, which leads to a greater overall participation level. This intuition is demonstrated in Fig. 4. From the third sub-figure in Fig. 3, we find that the uniform reward curve has several fluctuations suffering from the randomness of the social tie g_{ij} when network effects become stronger. Nevertheless, we can observe that the uniform reward still remains largely unchanged (around 50). This is different from the third subfigure in Fig. 2, where we cannot observe the fluctuations. The reason is that Fig. 2 illustrates the impacts of the number of MUs on the total offered reward from the CSP. Intuitively, the total reward increases when the number of participants increases. Thus, the slight fluctuations cannot be observed in Fig. 2 since the reward keeps increasing.

Then, to explore the impacts of social network effects on each specific participant, we investigate the optimal reward and resulting MUs' participation level with the number of 50 MUs, and we adopt the similar default parameters setting to that in above discussions. The adjacency matrix G is generated



Figure 4: A case illustration of distribution of normalized reward and participation level.

as follows:

$$\begin{cases} g_{i,i+1} = 0.2 \times \left(0.5 - \left(0.5 - \frac{i-1}{N} \right)^2 \right), & i \in [1, N-1]; \\ g_{i+1,i} = 0.2 \times \left(0.5 - \left(0.5 - \frac{i-1}{N} \right)^2 \right), & i \in [1, N-1]; \\ g_{i,j} = 0, & \text{otherwise.} \end{cases}$$
(30)

From Eq. (30), only participants who are adjacent in participant indexes (neighbours) can affect each other. From Fig. 4, we observe that the CSP offers each participant the same reward when it has no information about the value of matrix G. Given the reward from the CSP, the MUs have different participation level equilibrium as shown in Fig. 4, where we observe that the participation levels of the MUs are socially related to each other. In particular, the 27th MU is the most susceptible or influenced one in this network because it has the highest participation level given the same reward. On the contrary, the 1st and the 51st MUs are the most influential ones. Therefore, with the knowledge about the value of matrix G, the CSP is likely to offer more reward to the 1st and the 51st MUs and less to the 27th MU, under the discriminatory incentive mechanism. The reason is that the CSP tends to have the highest participation level from the participant with the lowest cost and thus have a greater revenue gain. However, under the uniform incentive mechanism, the CSP can offer only the same reward to the participants. As such, the CSP usually offers more reward and promotes the participants to attain higher participation level, but the incurred extra cost is also very high, which decreases the revenue of the CSP consequently.

B. Investigation on Bayesian Stackelberg game with incomplete information on social structure

Similar to the above discussions, we consider a group of N MUs. The in-degree and out-degree of MUs follow the normal distribution $\mathcal{N}(\bar{k}, \sigma_k^2)$ and $\mathcal{N}(\bar{k}, \sigma_l^2)$, respectively. The parameters are set as follows: $\gamma = 0.01$, $\bar{k} = 20$, $\sigma_k^2 = \sigma_l^2 = 10$, $\mu = 10$, s = 20, t = 0.05, c = 15, and N = 100.



Figure 5: The illustration of the optimal offered reward with respect to different in-degrees and out-degrees.

We first study the optimal offered reward in terms of different in-degrees and out-degrees, as illustrated in Fig. 5. Interestingly, we find that the optimal offered reward increases with the increase of in-degree, and increases with the decrease of out-degree. Recall that the MU's in-degree represents its influence and the out-degree represents its susceptibility. As the in-degree of the MU increases, this MU can encourage more other MUs due to the underlying social network effects. In a crowdsensing-based road traffic information sharing platform, we can treat the drivers in critical central paths as the MUs with higher in-degree, i.e., greater influence. The road information from these drivers plays a great role, i.e., the participation of these drivers can greatly promote the participation of others. Thus, in the presence of social network effects, the CSP tends to offer more reward to the MUs with the higher in-degree, since they potentially motivate more participation level of other MUs. On the contrary, for the MUs with the higher out-degree, the CSP has no incentive to offer more reward. The reason is that the MUs with the higher outdegree are more susceptible, and these MUs are potentially positively affected by others. Consequently, the CSP is able to offer less reward to save the cost.

Furthermore, we investigate the impacts of mean value of social network effects on the players of Bayesian Stackelberg game in Fig. 6. As expected, we observe that the optimal offered reward decreases when the mean value of social network effects increases. The reason is that as the mean value of social network effects become stronger, the MUs can motivate each other to have higher participation level due to the interdependent participation behaviors. The total utilities of MUs become greater, and thus the CSP tends to offer less reward to save the cost. Consequently, the CSP achieves the greater revenue gain. In addition, comparing different curves with different value of μ , we find that when μ , i.e., the equivalent monetary worth of MUs' participation level increases, the CSP tends to offer more reward as the participation incentive. The reason is that the CSP is more inclined to arouse the enthusiasms of MUs when the CSP can transform the participation level of MUs to more monetary



Figure 6: The impacts of mean value of social network effects.



Figure 7: The impacts of variance of the distribution of in/out-degree.

revenue efficiently. Therefore, to extract more surplus from MUs, the CSP offers more reward and thus achieves a greater revenue gain.

The

Figure 7 illustrates the impacts of variance of the distribution of social network effects on the players of Bayesian Stackelberg game. As the variance of social network effects decreases, the achieved revenue under uniform incentive mechanism is close to that under discriminatory incentive mechanism. The reason is that the heterogeneity of MUs is reduced when the value of variance decreases. We may consider an extreme case when the value of variance is zero, i.e., the MUs are homogeneous, the discriminatory incentive mechanism yields the same results as those of the uniform incentive mechanism. On the contrary, when the value of variance increases, the achieved revenue under the discriminatory incentive mechanism increases. The reason is that discriminatory incentive mechanism enables the CSP to exploit the different preference, i.e., parameters of utility function, for each of MUs, which leads to the decrease of total utilities of MUs and the increases of the revenue. Moreover, when the participation cost of MUs increases, the utility of MUs from participation is discounted. As such, the CSP tends to offer more reward to compensate the participation cost for improving their motivation. Meanwhile, the total utilities of MUs still decrease due to the increasing participation cost. Similarly, the revenue of the CSP decreases.

Lastly, both Figs. 6 and 7 demonstrate the fact that the discriminatory incentive mechanism performs better in terms of the achieved revenue compared with uniform incentive mechanism. The intuition is that, with the certain social structure information, the CSP can set different reward for different MUs, as verified by the Fig. 5. As such, the CSP can significantly encourage the greater participation level of MUs. In particular, the social structure information guides the CSP to extract more surplus from the MUs' participation, which results in the greater revenue gain.

In summary, we draw the following engineering insights:

- The network effects tremendously stimulate higher MU participation level, which leads to the greater total utilities of MUs as well as the greater revenue of the CSP.
- The discriminatory incentive mechanism yields the greater revenue of the CSP than that of the uniform incentive mechanism in both the complete and incomplete information scenarios.
- The revenue gap between the uniform and discriminatory incentive mechanisms depends on the variance of network effects.
- The CSP has an incentive to offer more reward to the influential MUs and less reward to the susceptible MUs, in order to promote a greater overall participation level.

VII. CONCLUSION

In this work, we have developed a two-stage Stackelberg game theoretic model, and obtained the equilibrium using backward induction. The Crowdsensing Service Provider (CSP) determines the incentive in the first stage, and the

Mobile Users (MUs) decide on their participation level in response to the observed incentive in the second stage. Taking the social (local) network effects among MUs into account, we have proposed two incentive mechanisms, i.e., discriminatory incentive and uniform incentive, where we have obtained the closed-form expression for optimal incentive. Moreover, we have formulated the Bayesian Stackelberg game to analyze the incentive mechanism, when the social network effects are uncertain. We have validated the existence and uniqueness of the Bayesian Stackelberg equilibrium by identifying the best response strategies of MUs. Performance evaluations have demonstrated that the network effects significantly improve the participation levels of MUs and the revenue of the CSP. Additionally, it has been confirmed that the social structure information helps the CSP to achieve greater revenue gain. The joint considerations of uncertainties of internal utility and social influence in the model are well worth studying in the future works.

APPENDIX

A. Proof of Theorem 3:

Proof. Similar to that in Section IV, we first apply the unique Bayesian Nash equilibrium of the follower game given in Eq. (28) into the objective function given in Eq. (18). Since r(k, l) = r for all MUs under the uniform incentive mechanism, we know that the unique participation level of the MU only depends on its out-degree k from Eq. (28), i.e., $x^*(k, l) = x^*(k)$. The expected revenue of the CSP given in (18) is then expressed as follows:

$$\Pi = \sum_{k \in D} P(k) \left((\mu s - r) x^*(k) - \mu t(x^*(k))^2 \right).$$
(31)

In particular, we have the mathematical transformations as shown in Eq. (32).

Then, we evaluate its first-order optimality condition with respect to the reward, and we have $\frac{\partial \Pi}{\partial r} = \frac{\partial \Pi}{\partial \frac{1+r-c}{1-\gamma \overline{k}}} \frac{\partial \frac{1+r-c}{1-\gamma \overline{k}}}{\partial r}$, which yields

$$\left(\left(\mu s + 1 - c \right) - 2 \left(1 - \gamma \overline{k} + \mu t + \mu t \gamma^2 \sigma_k^2 \right) \times \left(\frac{1 + r^* - c}{1 - \gamma \overline{k}} \right) \right) \frac{1}{1 - \gamma \overline{k}} = 0.$$
(33)

Thus, we can conclude that

$$1 + r^* - c = \frac{(\mu s + 1 - c) (1 - \gamma k)}{2 (1 - \gamma \overline{k} + \mu t + \mu t \gamma^2 \sigma_k^2)}.$$
 (34)

Therefore, the optimal uniform reward under Bayesian formulation is uniquely determined, which is given as follows:

$$r^{*} = c - 1 + \frac{(\mu s + 1 - c)(1 - \gamma k)}{2(1 - \gamma \overline{k} + \mu t + \mu t \gamma^{2} \sigma_{k}^{2})}.$$
 (35)

B. Proof of Theorem 4:

Proof. The CSP determines r(k, l) for the MU with outdegree k and in-degree l to maximize its expected revenue. The derivation of the optimal reward follows the similar steps discussed in Section IV. We have the expected revenue of the CSP, which is expressed as follows:

$$\Pi = \sum_{l \in D} \left(\sum_{k \in D} H(l) P(k) \left(\left(\mu s - r(k, l) \right) \left(1 + r(k, l) - c + \gamma k \frac{1 + \overline{r} - c}{1 - \gamma \overline{k}} \right) - \mu t \left(1 + r(k, l) - c + \gamma k \frac{1 + \overline{r} - c}{1 - \gamma \overline{k}} \right)^2 \right) \right).$$
(36)

By taking the first derivative with respect to r with any outdegree $m \in D$ and in-degree $n \in D$, we have Eq. (37). Since we know $\frac{\partial \Pi}{\partial r(m,n)} = 0$, we conclude that

$$0 = H(n)P(m)\left(-\left(1+r(m,n)-c+\gamma m\frac{1+\overline{r}-c}{1-\gamma\overline{k}}\right)+\mu s\right)$$
$$-r(m,n)-2\mu t\left(1+r(m,n)-c+\gamma m\frac{1+\overline{r}-c}{1-\gamma\overline{k}}\right)$$
$$+\gamma \frac{\overline{H}(n)P(m)}{1-\gamma\overline{k}}\sum_{l\in D}\sum_{k\in D}H(l)P(k)k\left(\mu s-r(k,l)-2\mu t\right)$$
$$\times\left(1+r(k,l)-c+\gamma k\frac{1+\overline{r}-c}{1-\gamma\overline{k}}\right).$$
(38)

With simple steps, we obtain the following expression

$$r^{*}(m,n) = \frac{1}{2(1+\mu t)} \left(c - 1 - \gamma m \frac{1+\overline{r}-c}{1-\gamma \overline{k}} + \mu s - 2\mu t \left(1 - c + \gamma m \frac{1+\overline{r}-c}{1-\gamma \overline{k}} \right) + \frac{\gamma n}{1-\gamma \overline{k}} \left(\mu s - 2\mu t \left(1-c \right) - \frac{2\mu t \gamma \left(1+\overline{r}-c \right) \left(\sigma _{k}^{2} + \overline{k}^{2} \right)}{\left(1-\gamma \overline{k} \right) \overline{k}} \right) - \gamma n \frac{1+2\mu t}{1-\gamma \overline{k}} \psi \right),$$
(39)

where

$$\psi = \sum_{l \in D} \sum_{k \in D} \frac{1}{\overline{k}} k H(l) P(k) r(k, l).$$
(40)

Moreover, based on the definition of \overline{r} , we have

$$\overline{r} = \sum_{m \in D} \sum_{n \in D} \overline{H}(n) P(m) r(m, n)$$

$$= \frac{1}{2(1+\mu t)} \left(c - 1 - \gamma \overline{k} \frac{1+\overline{r}-c}{1-\gamma \overline{k}} + \mu s - 2\mu t \left(1-c + \gamma \overline{k} \frac{1+\overline{r}-c}{1-\gamma \overline{k}}\right) + \frac{\gamma \overline{k}}{1-\gamma \overline{k}} \left(\mu s - 2\mu t \left(1-c\right) - \frac{2\mu t \gamma \left(1+\overline{r}-c\right) \left(\sigma_k^2 + \overline{k}^2\right)}{(1-\gamma \overline{k}) \overline{k}}\right) - \gamma \overline{k} \frac{1+2\mu t}{1-\gamma \overline{k}} \psi \right). \quad (41)$$

Thus, we have

$$\overline{r} = \frac{1}{2(1+\mu t)} \left(-\left(1-c+\gamma \overline{k}\frac{1+\overline{r}-c}{1-\gamma \overline{k}}\right) \left(\frac{2\mu t}{1-\gamma \overline{k}}+1\right) + \frac{\mu s}{1-\gamma \overline{k}} - \frac{2\mu t \gamma^2 \sigma_k^2 \left(1+\overline{r}-c\right)}{\left(1-\gamma \overline{k}\right)^2} \left(1-c+\gamma \overline{k}\frac{1+\overline{r}-c}{1-\gamma \overline{k}}\right) - \gamma \overline{k}\frac{1+2\mu t}{1-\gamma \overline{k}}\psi \right).$$

$$(42)$$

$$\Pi = (\mu s - r) \left(1 + r - c + \gamma \overline{k} \frac{1 + r - c}{1 - \gamma \overline{k}} \right) - \mu t \sum_{k \in D} P(k) \left(1 + r - c + \gamma \overline{k} \frac{1 + r - c}{1 - \gamma \overline{k}} \right)^2$$

$$= (\mu s - r) \frac{1 + r - c}{1 - \gamma \overline{k}} - \mu t (1 + r - c)^2 - 2\mu t (1 + r - c) \gamma \overline{k} \frac{1 + r - c}{1 - \gamma \overline{k}} - \mu t \left(\overline{k}^2 + \sigma_k^2 \right) \left(\gamma \frac{1 + r - c}{1 - \gamma \overline{k}} \right)^2$$

$$= (\mu s - r) \frac{1 + r - c}{1 - \gamma \overline{k}} - \mu t \gamma^2 \sigma_k^2 \left(\gamma \frac{1 + r - c}{1 - \gamma \overline{k}} \right)^2 - \left(\mu t (1 + r - c)^2 + 2\mu t (1 + r - c) \gamma \overline{k} \frac{1 + r - c}{1 - \gamma \overline{k}} + \mu t \overline{k}^2 \left(\gamma \frac{1 + r - c}{1 - \gamma \overline{k}} \right)^2 \right)$$

$$= (\mu s + 1 - c) \frac{1 + r - c}{1 - \gamma \overline{k}} - \left(1 - \gamma \overline{k} + \mu t + \mu t \gamma^2 \sigma_k^2 \right) \left(\frac{1 + r - c}{1 - \gamma \overline{k}} \right)^2$$
(32)

$$\frac{\partial \Pi}{\partial r(m,n)} = \frac{\partial}{\partial r(m,n)} H(n)P(m) \left(\left(\mu s - r(m,n)\right) \left(1 + r(m,n) - c + \gamma m \frac{1 + \overline{r} - c}{1 - \gamma \overline{k}} \right) - \mu t \left(1 + r(m,n) - c + \gamma m \frac{1 + \overline{r} - c}{1 - \gamma \overline{k}} \right)^2 \right) \\
+ \frac{\partial}{\partial r(m,n)} \sum_{l \neq n} \sum_{k \neq m} H(l)P(k) \left(\left(\mu s - r(k,l)\right) \left(1 + r(k,l) - c + \gamma k \frac{1 + \overline{r} - c}{1 - \gamma \overline{k}} \right) \right) \\
- \mu t \left(1 + r(k,l) - c + \gamma m \frac{1 + \overline{r} - c}{1 - \gamma \overline{k}} \right)^2 \right) \\
= H(n)P(m) \left(- \left(1 + r(m,n) - c + \gamma m \frac{1 + \overline{r} - c}{1 - \gamma \overline{k}} \right) + \left(1 + \gamma m \frac{\overline{H}(n)P(m)}{1 - \gamma \overline{k}} \right) (\mu s - r(m,n)) \right) \\
- 2\mu t \left(1 + r(m,n) - c + \gamma m \frac{1 + \overline{r} - c}{1 - \gamma \overline{k}} \right) \left(1 + \gamma m \frac{\overline{H}(n)P(m)}{1 - \gamma \overline{k}} \right) \right) + \frac{\partial}{\partial r(m,n)} \sum_{k \neq m} \sum_{l \neq n} H(l)P(k) \\
\times \left((\mu s - r(k,l)) \gamma k \frac{\overline{H}(n)P(m)}{1 - \gamma \overline{k}} - 2\mu t \gamma k \frac{\overline{H}(n)P(m)}{1 - \gamma \overline{k}} \left(1 + r(k,l) - c + \gamma m \frac{1 + \overline{r} - c}{1 - \gamma \overline{k}} \right) \right) \right) \tag{37}$$

Likewise, based on the definition of ψ given in Eq. (40), we know

$$\psi = \sum_{l \in D} \sum_{k \in D} \frac{1}{\overline{k}} k H(l) P(k) r(k, l).$$
(43)

Thus, we have Eq. (44).

The two expressions given in Eq. (42) and Eq. (44) together formulate a full rank linear equation system with two variables, i.e., \bar{r} and ψ . Thus, we can derive the closed-form expression for both \bar{r} and ψ [51]. In particular, the closed-form expression for the variable \bar{r} is obtained as Eq. (45), where $\rho = 1 + \frac{\gamma \bar{k}}{2(1+\mu t)-\gamma \bar{k}} \left(1 + \frac{2\mu t}{1-\gamma \bar{k}}\right) + \frac{\mu t \gamma^2 \sigma_k^2(1+\bar{r}-c)}{(1-\gamma \bar{k})^2(1+\mu t)} - \frac{(1+2\mu t)\gamma^2 \sigma_k^2}{2(2+2\mu t-\gamma \bar{k})(1-\gamma \bar{k})(1+\mu t)} \left(1 + \frac{2\mu t}{\gamma \bar{k}}\right)$. Likewise, the closed-form expression for the variable ψ can be derived through similar steps. Accordingly, r(m,n) can be obtained after plugging these two closed-form expressions into Eq. (39), and thus the solution of r(m,n) is unique. The proof is then completed.

REFERENCES

- [1] "Sensorly," http://www.sensorly.com/.
- [2] "Waze," https://www.waze.com/.
- [3] "Greengps," http://green-way.cs.illinois.edu/GreenGPS.html.
- [4] "Dietsensor," https://www.dietsensor.com/.
- [5] "Noisetube," http://www.noisetube.net/.
- [6] J. Surowiecki, M. P. Silverman et al., "The wisdom of crowds," American Journal of Physics, vol. 75, pp. 190–192, 2007.
- [7] A. Chakeri and L. Jaimes, "An incentive mechanism for crowdsensing markets with multiple crowdsourcers," *IEEE Internet of Things Journal*, 2017.
- [8] Y. Chen, B. Li and Q. Zhang, "Incentivizing crowdsourcing systems with network effects," in *Proceedings of IEEE INFOCOM*, San Francisco, CA, USA, April 2016.

- [9] M. Zhang, L. Yang, X. Gong and J. Zhang, "Privacy-preserving crowdsensing: Privacy valuation, network effect, and profit maximization," in *Proceedings of IEEE GLOBECOM*, Washington, DC, USA, December 2016.
- [10] D. Easley and J. Kleinberg, Networks, crowds, and markets: Reasoning about a highly connected world, Cambridge University Press, 2010.
- [11] O. Candogan, K. Bimpikis and A. Ozdaglar, "Optimal pricing in networks with externalities," *Operations Research*, vol. 60, no. 4, pp. 883–905, 2012.
- [12] Z. Xiong, S. Feng, D. Niyato, P. Wang and Y. Zhang, "Economic analysis of network effects on sponsored content: a hierarchical game theoretic approach," in *Proceedings of IEEE GLOBECOM*, Singapore, 2017.
- [13] X. Zhang, Z. Yang, W. Sun, Y. Liu, S. Tang, K. Xing and X. Mao, "Incentives for mobile crowd sensing: A survey," *IEEE Communications Surveys & Tutorials*, vol. 18, no. 1, pp. 54–67, 2016.
- [14] I. Koutsopoulos, "Optimal incentive-driven design of participatory sensing systems," in *Proceedings of IEEE INFOCOM*, Turin, Italy, April 2013.
- [15] J. Xu, J. Xiang and D. Yang, "Incentive mechanisms for time window dependent tasks in mobile crowdsensing," *IEEE Transactions on Wireless Communications*, vol. 14, no. 11, pp. 6353–6364, 2015.
- [16] K. Han, C. Zhang, J. Luo, M. Hu and B. Veeravalli, "Truthful scheduling mechanisms for powering mobile crowdsensing," *IEEE Transactions on Computers*, vol. 65, no. 1, pp. 294–307, 2016.
- [17] R. Zhou, Z. Li and C. Wu, "A truthful online mechanism for locationaware tasks in mobile crowd sensing," *IEEE Transactions on Mobile Computing*, 2017.
- [18] D. Yang, G. Xue, X. Fang and J. Tang, "Incentive mechanisms for crowdsensing: Crowdsourcing with smartphones," *IEEE/ACM Transactions on Networking (TON)*, vol. 24, no. 3, pp. 1732–1744, 2016.
- [19] Z. Zheng, F. Wu, X. Gao, H. Zhu, S. Tang and G. Chen, "A budget feasible incentive mechanism for weighted coverage maximization in mobile crowdsensing," *IEEE Transactions on Mobile Computing*, vol. 16, no. 9, pp. 2392–2407, 2017.
- [20] H. Wang, S. Guo, J. Cao and M. Guo, "Melody: a long-term dynamic quality-aware incentive mechanism for crowdsourcing," *IEEE Transactions on Parallel and Distributed Systems*, vol. 29, no. 4, pp. 901–914, 2018.
- [21] J. Lin, D. Yang, M. Li, J. Xu and G. Xue, "Frameworks for privacy-

$$\begin{split} \psi &= \frac{1}{2(1+\mu t)\bar{k}} \left(-\bar{k}\left(1-c\right) - \frac{\gamma\left(1+\bar{r}-c\right)\left(\sigma_{k}^{2}+\bar{k}^{2}\right)}{1-\gamma\bar{k}} + \mu s\bar{k} - 2\mu t \left(\left(1-c\right)\bar{k} + \frac{\gamma\left(1+\bar{r}-c\right)\left(\sigma_{k}^{2}+\bar{k}^{2}\right)}{1-\gamma\bar{k}}\right) \right. \\ &+ \frac{\gamma\bar{k}^{2}}{1-\gamma\bar{k}} \left(\mu s - \frac{2\mu t\gamma\left(1+\bar{r}-c\right)\left(\sigma_{k}^{2}+\bar{k}^{2}\right)}{(1-\gamma\bar{k})\bar{k}} - 2\mu t\left(1-c\right)\right) - \gamma\bar{k}^{2}\frac{1+2\mu t}{1-\gamma\bar{k}}\psi\right) \\ &= \frac{1}{2(1+\mu t)} \left(c-1 - \frac{\gamma\bar{k}\left(1+\bar{r}-c\right)\left(\sigma_{k}^{2}+\bar{k}^{2}\right)}{1-\gamma\bar{k}} + \mu s - 2\mu t \left(1-c + \frac{\gamma\bar{k}\left(1+\bar{r}-c\right)}{1-\gamma\bar{k}}\right)\right) \\ &+ \frac{\gamma\bar{k}}{1-\gamma\bar{k}} \left(\mu s - 2\mu t \left((1-c) + \frac{\gamma\bar{k}\left(1+\bar{r}-c\right)}{1-\gamma\bar{k}}\right) - \frac{2\mu t\gamma\left(1+\bar{r}-c\right)\sigma_{k}^{2}}{(1-\gamma\bar{k})\bar{k}}\right) \\ &- \frac{\gamma\sigma_{k}^{2}\left(1+\bar{r}-c\right)}{(1-\gamma\bar{k})\bar{k}} - \frac{2\mu t\gamma\sigma_{k}^{2}\left(1+\bar{r}-c\right)}{(1-\gamma\bar{k})\bar{k}} - \gamma\bar{k}\frac{1+2\mu t}{1-\gamma\bar{k}}\psi\right) \\ &= \frac{1}{2(1+\mu t)} \left(- \left(1-c + \frac{\gamma\bar{k}\left(1+\bar{r}-c\right)}{1-\gamma\bar{k}}\right) \left(1 + \frac{2\mu t}{1-\gamma\bar{k}}\right) + \frac{\mu s}{1-\gamma\bar{k}} - \frac{\gamma\sigma_{k}^{2}\left(1+\bar{r}-c\right)}{(1-\gamma\bar{k})\bar{k}} \left(1 + \frac{2\mu t}{1-\gamma\bar{k}}\psi\right) - \gamma\bar{k}\frac{1+2\mu t}{1-\gamma\bar{k}}\psi\right)$$
(44)

$$\overline{r} = \rho^{-1} \frac{\gamma \overline{k}}{2(1+\mu t) - \gamma \overline{k}} \left(-\left(1 - c + \frac{\gamma \overline{k} (1-c)}{1-\gamma \overline{k}}\right) \left(1 + \frac{2\mu t}{1-\gamma \overline{k}}\right) + \frac{\mu s}{\gamma \overline{k}} \right) - \frac{\mu t \gamma^2 \sigma_k^2 (1-c)}{(1-\gamma \overline{k})^2 (1+\mu t)} + \frac{(1+2\mu t)\gamma^2 \sigma_k^2 (1-c)}{2\left(2+2\mu t - \gamma \overline{k}\right)(1-\gamma \overline{k})(1+\mu t)} \left(1 + \frac{2\mu t}{\gamma \overline{k}}\right)$$
(45)

preserving mobile crowdsensing incentive mechanisms," *IEEE Transac*tions on Mobile Computing, 2017.

- [22] J. Lin, M. Li, D. Yang, G. Xue and J. Tang, "Sybil-proof incentive mechanisms for crowdsensing," in *Proceedings of IEEE INFOCOM*, Atlanta, GA, USA, May 2017.
- [23] Y. Zhan, Y. Xia, Y. Liu, F. Li and Y. Wang, "Incentive-aware timesensitive data collection in mobile opportunistic crowdsensing," *IEEE Transactions on Vehicular Technology*, vol. 66, no. 9, pp. 7849–7861, 2017.
- [24] X. Gan, Y. Li, W. Wang, L. Fu and X. Wang, "Social crowdsourcing to friends: An incentive mechanism for multi-resource sharing," *IEEE Journal on Selected Areas in Communications*, vol. 35, no. 3, pp. 795– 808, 2017.
- [25] R. Kawajiri, M. Shimosaka and H. Kashima, "Steered crowdsensing: incentive design towards quality-oriented place-centric crowdsensing," in *Proceedings of ACM Ubicomp*, Seattle, Washington, USA, September 2014.
- [26] D. Peng, F. Wu and G. Chen, "Pay as how well you do: A quality based incentive mechanism for crowdsensing," in *Proceedings of ACM Mobihoc*, Hangzhou, China, June 2015.
- [27] T. Luo, S. S. Kanhere, H.-P. Tan, F. Wu and H. Wu, "Crowdsourcing with tullock contests: A new perspective," in *Proceedings of IEEE INFOCOM*, Hong Kong, China, April 2015.
- [28] L. Duan, T. Kubo, K. Sugiyama, J. Huang, T. Hasegawa and J. Walrand, "Incentive mechanisms for smartphone collaboration in data acquisition and distributed computing," in *Proceedings of IEEE INFOCOM*, Orlando, Flodida, USA, March 2012.
- [29] K. Han, H. Huang and J. Luo, "Posted pricing for robust crowdsensing," in *Proceedings of ACM MobiHoc*, Paderborn, Germany, July 2016.
- [30] A. Chakeri and L. G. Jaimes, "An iterative incentive mechanism design for crowd sensing using best response dynamics," in *Proceedings of IEEE ICC*, Paris, France, May 2017.
- [31] Y. Zhan, Y. Xia and J. Zhang, "Incentive mechanism in platform-centric mobile crowdsensing: A one-to-many bargaining approach," *Computer Networks*, 2017.
- [32] J. Wang, M. Li, Y. He, H. Li, K. Xiao and C. Wang, "A blockchain based privacy-preserving incentive mechanism in crowdsensing applications," *IEEE Access*, vol. 6, pp. 17545–17556, 2018.
- [33] J. Nie, Z. Xiong, D. Niyato, P. Wang and J. Luo, "A socially-aware incentive mechanism for mobile crowdsensing service market," in *Proceedings of IEEE GLOBECOM*, Abu Dhabi, UAE, 2018.
- [34] Y. Li, C. A. Courcoubetis and L. Duan, "Dynamic routing for social information sharing," *IEEE Journal on Selected Areas in Communications*, vol. 35, no. 3, pp. 571–585, 2017.

- [35] E. Fehr and S. Gächter, "Fairness and retaliation: The economics of reciprocity," *Journal of economic perspectives*, vol. 14, no. 3, pp. 159– 181, 2000.
- [36] X. Chen, B. Proulx, X. Gong and J. Zhang, "Social trust and social reciprocity based cooperative d2d communications," in *Proceedings of* the ACM International Symposium on Mobile ad hoc Networking and Computing (Mobihoc). ACM, 2013, pp. 187–196.
- [37] J. Nie, J. Luo, Z. Xiong, D. Niyato and P. Wang, "A stackelberg game approach towards socially-aware incentive mechanisms for mobile crowdsensing (online report)," *arXiv preprint arXiv:1807.08412*, 2018.
- [38] X. Gong, L. Duan, X. Chen and J. Zhang, "When social network effect meets congestion effect in wireless networks: Data usage equilibrium and optimal pricing," *IEEE Journal on Selected Areas in Communications (JSAC)*, vol. 35, no. 2, pp. 449–462, 2017.
- [39] H. Moulin, "Dominance solvability and cournot stability," *Mathematical Social Sciences*, vol. 7, pp. 83–102, 1984.
- [40] D. P. Zhou, M. Roozbehani, M. A. Dahleh and C. J. Tomlin, "How peer effects influence energy consumption," in *Proceedings of IEEE CDC*, Melbourne, Australia, December 2017.
- [41] E. W. Weisstein, "Gershgorin circle theorem," 2003.
- [42] N. D. Duong, A. Madhukumar and D. Niyato, "Stackelberg bayesian game for power allocation in two-tier networks," *IEEE Transactions on Vehicular Technology*, vol. 65, no. 4, pp. 2341–2354, 2016.
- [43] W. Chen, P. Lu, X. Sun, B. Tang, Y. Wang and Z. A. Zhu, "Optimal pricing in social networks with incomplete information," in *International Workshop on Internet and Network Economics*. Springer, 2011, pp. 49– 60.
- [44] F. Bloch and N. Quérou, "Pricing in social networks," Games and economic behavior, vol. 80, pp. 243–261, 2013.
- [45] Y. Zhang and Y.-J. Chen, "Optimal nonlinear pricing in social networks under asymmetric network information," forthcoming.
- [46] I. P. Fainmesser and A. Galeotti, "Pricing network effects: Competition," SSRN Electronic Journal, early access, 2017.
- [47] M. Belhaj and F. Deroian, "The value of network information: Assortative mixing makes the difference," AMSE Working Papers, early access, 2017.
- [48] M. Newman, Networks, Oxford university press, 2018.
- [49] E. L. Glaeser and J. Scheinkman, "Non-market interactions," Tech. Rep., National Bureau of Economic Research, 2000.
- [50] Z. Han, D. Niyato, W. Saad, T. Baar and A. Hjrungnes, *Game theory in wireless and communication networks: theory, models, and applications*, Cambridge University Press, 2012.
- [51] T. Kailath, *Linear systems*, vol. 156, Prentice-Hall Englewood Cliffs, NJ, 1980.



Jiangtian Nie (S'18) received her B.Eng degree with honors in Electronics and Information Engineering from Huazhong University of Science and Technology, Wuhan, China, in 2016. She is currently working towards the Ph.D. degree with ERI@N in the Interdisciplinary Graduate School, Nanyang Technological University, Singapore. Her research interests include incentive mechanism design in crowdsensing and game theory.



Jun Luo (M'09-SM'18) received his BS and MS degrees in Electrical Engineering from Tsinghua University, China, and the Ph.D. degree in Computer Science from EPFL (Swiss Federal Institute of Technology in Lausanne), Lausanne, Switzerland. He is currently an Associate Professor at School of Computer Engineering, Nanyang Technological University. His research interests include mobile computing and applied operations research, as well as network security.



Zehui Xiong (S'17) received his B.Eng degree with honors in Telecommunication Engineering from Huazhong University of Science and Technology, Wuhan, China, in 2016. He is currently working towards his Ph.D. degree in the School of Computer Science and Engineering, Nanyang Technological University, Singapore. His research interests include network economics, game theory for resource management, market models and pricing.



Dusit Niyato (M'09-SM'15-F'17) is currently a full professor in the School of Computer Science and Engineering, at Nanyang Technological University, Singapore. He received B.Eng. from King Mongkuts Institute of Technology Ladkrabang (KMITL), Thailand in 1999 and Ph.D. in Electrical and Computer Engineering from the University of Manitoba, Canada in 2008. His research interests are in the area of energy harvesting for wireless communication, Internet of Things (IoT) and sensor networks.



Ping Wang (M'08-SM'15) received the PhD degree in electrical engineering from University of Waterloo, Canada, in 2008. Currently, she is an Associate Professor in the Department of Electrical Engineering and Computer Science, York University, Canada. Her current research interests include resource allocation in multimedia wireless networks, cloud computing, and smart grid. She was a corscipient of the Best Paper Award from IEEE Wireless Communications and Networking Conference (WCNC) 2012 and IEEE International Conference

on Communications (ICC) 2007. She is an Editor of IEEE Transactions on Wireless Communications, EURASIP Journal on Wireless Communications and Networking, and International Journal of Ultra Wideband Communications and Systems.